Author's response to reviews

Title: Asymptomatic Bacteriuria, Antibiotic Use, and Suspected Urinary Tract Infections in Four Nursing Homes

Authors:

Charles D Phillips (Phillipscd@srph.tamhsc.edu)
Omolola Adepoju (adepoju@srph.tamhsc.edu)
Nimalie Stone (nstone@cdc.gov)
Darcy K Moudouni (dkmoudouni@srph.tamhsc.edu)
Obioma B Nwaiwu (nwaiwu@srph.tamhsc.edu)
Hongwei Zhao (zhao@srph.tamhsc.edu)
Elizabeth Frentzel (Efrentzel@air.org)
David Mehr (mehrd@health.missouri.edu)
Steven Garfinkel (sgarfinkel@air.org)

Version: 6 Date: 12 October 2012

Author's response to reviews: see over
Asymptomatic Bacteriuria, Antibiotic Use, and Suspected Urinary Tract Infections in Four Nursing Homes

RESPONSE TO COMMENT BY ASSOCIATE EDITOR

COMMENT: Table 5 is confusing and does not add any knowledge and does not use adequate comparison methods. The authors are dividing each variable into low and high deciding beforehand that two are low and two are high. they should either do a proper statistical analyses to compare characteristics between centres or completely delete Table 5 and the last paragraph that goes with it."

RESPONSE: The manuscript has been revised and Table 5, the last paragraph, and all texts associated with Table 5 deleted.